Author's response to reviews

Title: Intubating conditions and side effects of propofol, remifentanil and sevoflurane compared with propofol, remifentanil and rocuronium: a randomised, prospective, clinical trial

Authors:

Thomas Mencke (thomas.mencke@googlemail.com)
Refa Maria Jacobs (r-m.jacobs@web.de)
Susann Machmueller (susann.machmueller@uni-rostock.de)
Martin Sauer (martinsauer.sauer@googlemail.com)
Christine Heidecke (christine.heidecke@gmx.de)
Anja Kallert (anja.kallert@med.uni-rostock.de)
Hans Wilhelm Pau (hans-wilhelm.pau@med.uni-rostock.de)
Gabriele Noeldge-Schomburg (gabriele.noeldge-schomburg@uni-rostock.de)
Attila Ovari (atiova@gmail.com)

Version: 2 Date: 17 January 2014

Author's response to reviews: see over
Dear Editor of BMC Anesthesiology,

we would like to submit the manuscript *Intubating conditions and side effects of propofol, remifentanil and sevoflurane compared with propofol, remifentanil and rocuronium: a randomised, prospective, clinical trial* to *BMC Anesthesiology* as a research article.

We added the Acknowledgements’ section to the manuscript.

With many thanks for your interest,

*Thomas Mencke, MD*